Author's response to reviews

Title: Randomized Controlled Trial to Evaluate Tooth Stain Reduction with Nicotine Replacement Gum during a Smoking Cessation Program

Authors:

Helen Whelton (h.whelton@ucc.ie)
Rose Kingston (r.kingston@ucc.ie)
Denis O'Mullane (d.omullane@ucc.ie)
Frederick Nilsson (ekruse@its.jnj.com)

Version: 2 Date: 18 November 2011

Author's response to reviews:

The registration of the trial has been addressed. Although the trial was registered with EUDRA CT, this was not on the list of accepted registries. The trial is now registered with Clinicaltrials.gov which is on the accepted list. New registration details have been entered.

Greater detail on the contribution of the funding company McNeil AB has been provided.